Author's response to reviews

Title: The Role of Institutions in the Effectiveness of Treatment in the Ghanaian Health Sector

Authors:

Eugenia Amporfu (eamporfu@gmail.com)
Justice Nonvignon (justicenon@gmail.com)

Version: 2 Date: 14 September 2013

Author's response to reviews: see over
Dear Editor,

Please find below the changes made to manuscript MS: 3079102361076840 entitled: The Role of Institutions in the Effectiveness of Treatment in the Ghanaian Health Sector.

1. In the methods section I inserted the organization that provided ethical approval for the survey and that informed consent forms were given to respondents to sign.
2. Four keywords were included after the abstract
3. The Data description and regression results sections were merged under the Results and Discussion section.
4. All tables were moved to the end of the paper, after the references
5. Competing interest was added after the conclusion
6. Authors contributions was included after the competing interest.
7. Acknowledgment was added.

Track changes in MS word was used and so the changes can easily be found in the manuscript.

Please do not hesitate to contact me should you have any problem.

Yours faithfully,

Eugenia Amporfu